Author's response to reviews

Title: Social Networks and Implementation of Evidence-Based Practices in Public Youth-Serving Systems: A Mixed Methods Study

Authors:

Lawrence A Palinkas (palinkas@usc.edu)
Ian W Holloway (ihollowa@usc.edu)
Eric Rice (ericr@usc.edu)
Dahlia Fuentes (dfuentes@usc.edu)
Qiaobing Wu (qiaobing.wu@gmail.com)
Patricia Chamberlain (pattic@oslc.org)

Version: 8 Date: 29 September 2011

Author's response to reviews:

Dear Independent Editorial Production Team:

At your request, we have added a sentence in the Competing Interests section indicating that "All other authors declare no competing interests." We have also expanded the size of the figure so that it contains as small a margin as is possible.

Sincerely,

Lawrence A. Palinkas, Ph.D.